Author's response to reviews

Title: Health-service performance of TB treatment for indigenous and non-indigenous populations in Brazil: A cross-sectional study

Authors:

Everton F Lemos (tonufms@gmail.com)
Aline M S Alves (aline.msa@hotmail.com)
Giovana de C Oliveira (giovanacastro4@gmail.com)
Marcella P Rodrigues (donaestoica@hotmail.com)
Natália D G Martins (nathy_dgm@yahoo.com.br)
Julio Croda (juliocroda@gmail.com)

Version: 4
Date: 2 March 2014

Author's response to reviews: see over
Christopher Morrey  
BioMed Central Editorial  

Dourados, March 2, 2014

Dear Editor:  
We appreciate your consideration of our work and accepted the manuscript for publication in BMC Health Services Research. We are submitting the revised version of the manuscript, “Health-service performance of TB treatment for indigenous and non-indigenous populations in Brazil: A cross-sectional study” (MS: 2018415531852025). We have addressed the reviewers’ comments in the responses below.

Reviewers' Critiques:

Reviewer: 1
None

Reviewer: 2
Minor essential revisions

Line 184-185: Please amend reference (i.e. remove "Villa et al 2011")

Line 250: Please amend acronym to DOT for consistency

Author’s reply: Revised as suggested.

Sincerely,
Julio Croda, MD, PhD  
Faculty of Health Sciences  
Federal University of Grande Dourados  
Rodovia Dourados - Itaúm, km 12, Dourados, Mato Grosso do Sul 79804-970, Brazil  
Phone: 55-67-3410-2321 Fax: 55-67-3410-2320  
juliocroda@ufgd.edu.br; juliocroda@gmail.com